Author's response to reviews

Title: Electrical short-circuit in beta-cells from a patient with non-insulinoma pancreatogenous hypoglycemic syndrome (NIPHS): A case report

Authors:

Robert Bränström (robert.branstrom@ki.se)
Erik Berglund (Erik.Berglund@ki.se)
Pontus Curman (pontus.curman@sodersjukhuset.se)
Lars Forsberg (larsforsberg65@gmail.com)
Anders Höög (anders.hoog@karolinska.se)
Lars Grimelius (lars.grimelius@genpat.uu.se)
Per-Olof Berggren (Per-Olof.Berggren@ki.se)
Per Mattsson (Per.Mattsson@karolinska.se)
Per Hellman (per.hellman@surgsci.uu.se)
Lisa Juntti-Berggren (Lisa.Juntti-Berggren@ki.se)

Version: 2 Date: 5 December 2009

Author's response to reviews:

Dear Editor,

As requested we have now added a short introduction setting the focus on the background and clinical problem. Information regarding the patient origin as also been added.

Yours sincerely,

Robert Bränström
M.D., Ph.D., Assoc professor
Section of Endocrine Surgery
Department of Molecular Medicine and Surgery
Karolinska Institutet, Stockholm
Sweden